Author's response to reviews

Title: Systematic review of safety checklists for use by medical care teams in acute hospital settings - limited evidence of effectiveness

Authors:

Henry CH Ko (henry.ko@ctc.usyd.edu.au)
Tari J Turner (tari.turner@monash.edu)
Monica A Finnigan (monica.finnigan@southernhealth.org.au)

Version: 8 Date: 28 August 2011

Author's response to reviews:

Dear BMC Series Editorial Production Team,

I have deleted a repeated sentence about no meta-analysis being done in the methods section of the article. There is now only one sentence stating that no meta-analysis was done. Apart from that no other changes were made. I am happy for the article to be published. Thank you.

Kind Regards,

Henry Ko.